Author's response to reviews

Title: Effects of light-emitting diodes on muscle fatigue and exercise tolerance in patients with COPD: a randomized clinical trial

Authors:

Eduardo F Miranda (foschinimedu@ig.com.br)
Ernesto CPL Junior (ernesto.leal.junior@uninove.br)
Paulo H Marchetti (dr.pmarchetti@gmail.com)
Simone Dal Corso (simonecd@uninove.br)

Version: 2 Date: 30 July 2012

Author's response to reviews: see over
July 30th, 2012

To the Editor in Chief
Doug Altman, University of Oxford

Re: Effects of light-emitting diodes on muscle fatigue and exercise tolerance in patients with COPD: a randomized clinical trial

Dear Professor,

The study protocol was approved by the Institutional Ethics Committee of our institution (Nove de Julho University – Protocol 11/451953) and also has been registered in the Clinical Trial.gov (NCT 01448564). And we modified the manuscript as suggested.

Yours sincerely,

Simone Dal Corso, PT, PhD
Professor of Rehabilitation Science Master’s Program
Nove de Julho University